Author’s response to reviews

Title: Environmental risk factors for dementia: a systematic review

Authors:

Lewis Killin (lewiskillin@nhs.net)
John Starr (jstarr@staffmail.ed.ac.uk)
Ivy Shiue (ivy.shiue@northumbria.ac.uk)
Tom Russ (tc.russ@ed.ac.uk)

Version: 2 Date: 26 Aug 2016

Author’s response to reviews:

26th August 2016

Dear Professor Gu,

Research article: Environmental risk factors for dementia: a systematic review

Thank you for sending your comment. We have made the requested change and have marked it in the manuscript in red text.

We look forward to hearing from you,

Dr Tom Russ PhD MRCPsych